Author's response to reviews

Title: Response of lymphocyte subsets and cytokines to Shenyang prescription in Sprague-Dawley rats with tongue squamous cell carcinomas induced by 4NQO

Authors:

Canhua Jiang (canhua-j@sohu.com)
Dongxia Ye (yedongxia122@hotmail.com)
Weiliu Qiu (qwl@omschina.org.cn)
Xiuli Zhang (zhxl59@yahoo.com.cn)
Zhiyuan Zhang (zhzhy@omschina.org.cn)
Di He (enamel1981@hotmail.com)
Ping Zhang (pingzhang73@hotmail.com)
Wantao Chen (chenwantao2002@hotmail.com)

Version: 6 Date: 2 March 2007

Author's response to reviews:

Dear Editors

Thank you for accepting our paper Response of lymphocyte subsets and cytokines to Shenyang prescription in Sprague-Dawley rats with tongue squamous cell carcinomas induced by 4NQO (MS: 1728714112121869) for being published. We have revised our manuscript as suggested by reviewer point by point.

1. Added a Conclusions section after the Discussion.
2. Removed the underlining from our manuscript.
3. Added an affiliation for some authors.

We greatly appreciate your reception of our work again.

Sincerely,

Wantao Chen,